Author's response to reviews

Title: Does Chinese culture influence psychosocial factors for heroin use among young adolescents in China? A cross-sectional study

Authors:

   Hongjie Liu (hliu@vcu.edu)
   Jian Li (jli4@vcu.edu)
   Zhouping Lu (lzp0427047@126.com)
   Wei Liu (lw_gx@126.com)
   Zhiyong Zhang (rpazz@163.com)

Version: 3 Date: 17 September 2010

Author's response to reviews: see over
September 17, 2010

Editors,

BMC Public Health

Dear Editors:

Thank you for accepting our manuscript for publication. As per your request, we have made the following minor changes:

1. A heading titled ‘Conclusions’ has been inserted.
2. In the part of the authors’ contributions, full names have been changed to initials.
3. The color of text fonts has been changed to black.
4. We have checked for typos and made corrections.

Thank you again for your acceptance of our manuscript for publication.

Sincerely yours,

Hongjie

Hongjie Liu, Ph.D., M.S.
Associate Professor
Department of Epidemiology and Community Health
Virginia Commonwealth University School of Medicine
830 East Main Street
P.O. Box 980212
Richmond, VA 23298
Tel: 804-628-2517, Fax: 804- 828-9773
e-mail: hliu@vcu.edu